@AGUPUBLICATIONS

Journal of Geophysical Research: Solid Earth

RESEARCH ARTICLE

10.1002/2016JB013599

Key Points:

- Fossil anisotropy preserved in the upper lithosphere suggests that cratonic blocks sustained plate-scale collision in the Archean
- Lower cratonic lithosphere is weakly anisotropic, suggesting postassembly lithospheric growth in a slow convection regime
- Lowermost lithosphere is highly heterogeneous and anisotropic, and transition to underlying asthenosphere is complex

Supporting Information:

- Supporting Information S1
- Data Set S1
- Data Set S2
- Data Set S3
- Data Set S4

Correspondence to:

L. Petrescu, l.petrescu13@imperial.ac.uk

Citation:

Petrescu, L., F. Darbyshire, I. Bastow, E. Totten, and A. Gilligan (2017), Seismic anisotropy of Precambrian lithosphere: Insights from Rayleigh wave tomography of the eastern Superior Craton, J. Geophys. Res. Solid Earth, 122, 3754–3775, doi:10.1002/2016JB013599.

Received 28 SEP 2016 Accepted 24 APR 2017 Accepted article online 26 APR 2017 Published online 13 MAY 2017

©2017. American Geophysical Union. All Rights Reserved.

Seismic anisotropy of Precambrian lithosphere: Insights from Rayleigh wave tomography of the eastern Superior Craton

Laura Petrescu^{1,2}, Fiona Darbyshire³, Ian Bastow¹, Eoghan Totten^{1,4}, and Amy Gilligan^{1,5}

¹Department of Earth Science and Engineering, Imperial College London, London, UK, ²National Institute for Earth Physics, Magurele, Romania, ³Centre de recherche GEOTOP, Université du Québec à Montréal, Montréal, Quebec, Canada, ⁴Now at Department of Earth Sciences, University of Oxford, Oxford, UK, ⁵Now at School of Geosciences, University of Aberdeen, Aberdeen, UK

JGR

Abstract The thick, seismically fast lithospheric keels underlying continental cores (cratons) are thought to have formed in the Precambrian and resisted subsequent tectonic destruction. A consensus is emerging from a variety of disciplines that keels are vertically stratified, but the processes that led to their development remain uncertain. Eastern Canada is a natural laboratory to study Precambrian lithospheric formation and evolution. It comprises the largest Archean craton in the world, the Superior Craton, surrounded by multiple Proterozoic orogenic belts. To investigate its lithospheric structure, we construct a frequency-dependent anisotropic seismic model of the region using Rayleigh waves from teleseismic earthquakes recorded at broadband seismic stations across eastern Canada. The joint interpretation of phase velocity heterogeneity and azimuthal anisotropy patterns reveals a seismically fast and anisotropically complex Superior Craton. The upper lithosphere records fossilized Archean tectonic deformation: anisotropic patterns align with the orientation of the main tectonic boundaries at periods ≤110 s. This implies that cratonic blocks were strong enough to sustain plate-scale deformation during collision at 2.5 Ga. Cratonic lithosphere with fossil anisotropy partially extends beneath adjacent Proterozoic belts. At periods sensitive to the lower lithosphere, we detect fast, more homogenous, and weakly anisotropic material, documenting postassembly lithospheric growth, possibly in a slow or stagnant convection regime. A heterogeneous, anisotropic transitional zone may also be present at the base of the keel. The detection of multiple lithospheric fabrics at different periods with distinct tectonic origins supports growing evidence that cratonization processes may be episodic and are not exclusively an Archean phenomenon.

Plain Language Summary The roots of the oldest parts of the continents are unusually thick (~250 km) and preserve structures dating back to the Earth's most ancient geological period, >2.5 billion years ago. The structure, formation, and preservation of these thick continental roots are still debated. Eastern Canada has a geological record stretching back over more than 3 billion years, providing an excellent opportunity to study the Earth's early history and the formation and evolution of continents. To investigate the deep structure beneath this region, we use earthquake energy recorded at seismic stations across eastern Canada to build an image of the continental interior. Our results show that the ancient continental root is made up of at least two layers, each with a different formation mechanism. The upper layer preserves the signature of continental deformation older than 2.5 billion years, whereas the lower layer properties suggest later downward growth of the root above a slow-moving mantle. The results support the idea that the thick continental roots formed episodically and across multiple geological periods.

1. Introduction

Cratons are the geologically distinct Archean (>2.5 Ga) cores of the continents that have survived multiple Wilson cycles. They are typically underlain by thick (\leq 300 km), seismically fast [e.g., *Schaeffer and Lebedev*, 2013] lithospheric roots (keels) that may be highly depleted in Ca, Al, Fe, and Mg [*Griffin et al.*, 2003]. Some keels have been found to comprise fast wave speed, chemically depleted upper layers, underlain by a cold but compositionally less distinct thermal boundary layer [*Yuan and Romanowicz*, 2010].

The deep structure of cratons worldwide has been analyzed using a range of seismic imaging methods and through kimberlite-borne mantle xenoliths. Broadband seismology studies detected multiple intralithospheric seismic discontinuities [e.g., *Romanowicz*, 2009; *Abt et al.*, 2010; *Wirth and Long*, 2014] that may represent the bottom of the chemically depleted layers and younger cratonic roots. Mantle xenoliths in some cratons (e.g., the Slave Craton in NW Canada and the North Atlantic Craton in Greenland [*Griffin et al.*, 1999; *Kopylova and Russell*, 2000; *Bizzarro and Stevenson*, 2003]) record compositional stratification, although this is not evident in other cratons such as the Siberian or the Kaapvaal Cratons [e.g., *Boyd et al.*, 1997]. The existence of multiple anisotropic layers has also been reported, with a change from lithospheric fabrics to present-day mantle flow beneath several stable continental regions, e.g., North America [*Darbyshire and Lebedev*, 2009; *Yuan and Romanowicz*, 2010], South Africa [e.g., *Adam and Lebedev*, 2012], northern Europe [e.g., *Plomerová and Babuska*, 2010], and Australia [e.g., *Fichtner et al.*, 2010].

The tectonic processes that led to keel development remain uncertain. A range of contrasting generation mechanisms has been proposed, including mantle plume melt extraction [e.g., *Arndt et al.*, 2009], catastrophic delamination events [*Bédard*, 2006], and multiple island arc accretion episodes [e.g., *Miller and Eaton*, 2010]. The origin of the intracratonic layers and discontinuities has also been attributed to metasomatism [e.g., *Savage and Silver*, 2008], partial melt [e.g., *Thybo*, 2006], elastically accommodated grain boundary sliding [*Karato*, 2012], decrease in Mg content with depth [e.g., *Yuan and Romanowicz*, 2010], and stacked subduction of noneclogitized oceanic crust [e.g., *Bostock et al.*, 2010]. Key to discriminating these hypotheses and to understanding the nature and relative timing of cratonization processes is an improved detailed knowledge of the lithospheric structure beneath Archean continental regions and its subsequent deformation history.

Eastern Canada is home to one of the largest and best preserved Archean cratons in the world, the Superior Craton, thought to have been assembled by 2.5 Ga [*Hoffman*, 1989]. Its edges experienced collisional orogenies and Andean-style subduction throughout the Proterozoic [e.g., *Rivers*, 1997] and witnessed the closure of the Rheic and lapetus Oceans during the Phanerozoic [*van Staal and Barr*, 2012]. Eastern Canada thus provides an excellent opportunity to gain fundamental insights about the types of tectonic processes that formed the cratonic mantle and subsequently modified it.

To investigate the geological structures beneath eastern Canada, we use earthquake data to construct a tomographic image of the subsurface in period domain. Using the dispersion properties of surface waves, we estimate the variation of isotropic and anisotropic components of the Rayleigh wave phase velocity [*Deschamps et al.*, 2008], between 20 and 220 s period, broadly sensitive to depths of 30–350 km. Variations in phase velocity are used to map out the extent of the cratonic core beneath the study region. Measurements of seismic anisotropy allow us to observe fabrics from episodes of tectonic deformation that affected various parts of the cratonic keel. Our results provide evidence for the multistage mechanisms that participated in the formation and evolution through space and time of a layered cratonic mantle.

2. Tectonic Setting

The geology of Eastern Canada documents tectonic events dating back to the Archean eon (>2.5 Ga). It includes one of the largest cratons in the world, the Superior Craton (Figure 1), which was part of the Laurentia paleocontinent. The Superior Craton is composed of several microcratonic blocks, whose tectonic assemblage was completed by ~2.5 Ga [*Hoffman*, 1989]. The dominant direction of Archean surface tectonic boundaries changes from E-W in the south and west of the province to N-S in the northeast (Figure 1).

The edges of the Superior Craton experienced multiple supercontinent and Wilson cycles, including the Paleoproterozoic Trans-Hudson orogeny to the north and west and the New Quebec orogeny to the northeast, respectively (Figure 1). The latter is part of the Eastern Churchill Province [*Wardle et al.*, 1990], comprising an Archean core trapped between two Paleoproterozoic N-S trending orogenic belts, the New Quebec and Torngat orogens.

A long-lived Andean-style paleosubduction system is thought to have dominated the eastern cratonic margins of the Superior between ~1.7 Ga and ~1.2 Ga [*Rivers*, 1997]. Between 1090 Ma and 980 Ma, the Superior Craton was involved in a Himalayan-scale collision with Amazonia [*Li et al.*, 2008] as part of one of the most important supercontinent cycles in the Proterozoic. The Grenville Province is the remnant of this Proterozoic orogeny and comprises rocks that were originally part of the Archean Superior Craton [*Rivers*, 2009].



Figure 1. (a) The tectonic framework of eastern Canada. Colors mark the age of the surface geology: Archean (pink), Proterozoic (green), and Phanerozoic (purple). Light purple indicates a region of the Superior Craton covered by Phanerozoic sediments. Black lines mark the major tectonic boundaries. Thin black lines are surface geological boundaries. The dashed line marks the probable western limit of the Superior Craton, beneath Hudson Bay. The yellow dashed line marks the Great Meteor hot spot track. NQ, New Quebec Orogen; TO, Torngat Orogen; THO, Trans-Hudson Orogen. (b) Seismic ray paths (green lines) connecting pairs of seismic stations in eastern Canada. Earthquake data recorded at these stations were used to compute Rayleigh wave phase velocity dispersion curves. Inset: earthquakes used to obtain dispersion curves.

Southeast of the Grenville Province, the Appalachian orogenic belt records the closure of the lapetus and Rheic Oceans between 480 Ma and 260 Ma [*van Staal and Barr*, 2012].

Between 160 Ma and 80 Ma, the North American plate drifted over the Great Meteor hot spot, with surface geological signatures such as kimberlite pipes in the Superior Craton and a track of igneous intrusions in the Grenville Province [*Sleep*, 1990].

3. Method and Data

3.1. Rayleigh Wave Dispersion

Rayleigh waves travel with phase velocities that are frequency dependent and are a powerful tool to study upper mantle structure. By cross-correlating surface waveforms simultaneously recorded at pairs of seismic stations, it is possible to estimate the path-averaged Rayleigh wave phase velocity between them as a function of wave period [*Meier et al.*, 2004].

We processed seismic records from 3218 teleseismic earthquakes with $M_w > 5.5$ between 2009 and 2014 that were recorded at 48 permanent and temporary broadband seismic stations in eastern Canada, Greenland, southern Baffin Island, and the northeast U.S. (Figure 1). Data from stations outside of the region of interest widens the aperture of the network, improving the resolution of our model at the edges. Of 1128 possible station pairs, ~500 recorded earthquakes along great-circle paths within $\pm 5^{\circ}$ of the interstation back azimuth and were thus usable for analysis.

The recorded displacement is approximated with a Jordan-Wenzel-Kramera-Brillouin (JWKB) response to a moment tensor source on an isotropic, spherically symmetric Earth model [*Snieder and Nolet*, 1987]. The vertical displacement seismograms recorded at pairs of seismic stations are shown in Figure 2, along with their amplitude as a function of time and frequency, to emphasize the frequency range for which fundamental mode Rayleigh waves can be used in the two-station method analysis. Strong scattering influences the 0.6–0.1 Hz frequency range (10–16 s period). The cross-correlation function is plotted in Figure 2e along with its frequency-time representation (Figure 2f). A frequency-dependent Gaussian bandpass filter was applied to enhance the signal-to-noise ratio and minimize noise and interferences with scattered waves. A subsequent time domain frequency-dependent Gaussian filter, centered around the peak amplitude time, was applied to attenuate the side lobe effects arising from fundamental mode interference with higher modes. The cross-correlation function is then transferred to the frequency domain, and its complex phase is used



Figure 2. The interactive cross-correlation procedure for constructing a dispersion curve. The vertical displacement seismograms of an earthquake as recorded at stations (a) ACCN and (c) SUNO. (b,d) The corresponding spectrograms of the recorded seismograms. (e) The time series cross correlation of the two plotted seismograms. (f) The corresponding spectrogram of the cross-correlation function. (g) Rayleigh wave phase velocity dispersion curves. Red lines are the array of curves resulting from the calculation of phase velocity from the phase of the cross-correlation function. Black dashed line is a theoretical dispersion curve calculated from the ak135 model. (h) The dispersion curves interactively selected from previous cross correlations of recorded seismic energy of earthquakes coming from the direction of station ACCN (black lines) or SUNO (green lines).

to estimate the frequency-dependent phase velocity, $c(\omega)$, where ω is the angular frequency (see *Meier et al.* [2004] and *Sato* [1955] for a comprehensive description of the method and related equations):

$$(\omega) = \frac{\omega(\Delta_1 - \Delta_2)}{\arctan\{\Re[\phi(\omega)]/\Im[\phi(\omega)]\} + 2n\pi},\tag{1}$$

where Δ_1 and Δ_2 are the epicentral distances of the two stations and ϕ is the cross-correlation function. The $2n\pi$ term in the denominator emerges from the intrinsic ambiguity of the arctan function. The frequency-dependent phase velocity solution is thus nonunique, comprising a discrete set of dispersion curves, one for each value of n (Figure 2g). The most plausible value of n or, equivalently, the most likely dispersion curve, is recovered by visually inspecting the solution space and manually selecting the portion of the dispersion curve most similar to a theoretical global reference curve (Figure 2g). Cross-correlation procedures suffer from cycle-skipping problems, which may arise if the phase difference between the two signals is larger than half a period of the dominant wavelet. To minimize cycle skipping, we perform the computation of the phase starting from the longest period. Incorrect phase unwrapping will result in a discontinuity in the dispersion curves, which is easily detectable in the manual selection step. The final dispersion curve is the average of the selected solutions for all recorded earthquakes at the two seismic stations. Seismic diffraction and fundamental mode Rayleigh wave interference with higher modes can introduce irregularities and roughness in the measured curve [e.g., Legendre et al., 2014]. However, these effects are frequency dependent and are smoothed in the manual selection and averaging steps (Figures 2g and 2h). The resulting dispersion curve is a path-averaged phase velocity variation of the Rayleigh waves sampling the structures between the two stations. Using this method, we estimated 285 interstation dispersion curves (Figure 1) for the period range 20-220 s from ~180,000 seismograms. The number of curves accurately retrieved for each period varied between 189 and 276.

С

We minimize potential bias from off-great-circle propagation [*Pedersen*, 2006] by combining dispersion measurements from multiple earthquakes at different epicentral distances and reciprocal paths for each station pair (see Appendix A for more information).

3.2. Seismic Tomography

Each dispersion curve yields an average phase velocity along the interstation path it was estimated for, as a function of period. The total phase velocity anomaly $\delta \overline{C_i}$ can be written as the integral of all the local anomalies $(\delta C(\varphi, \theta))$ at each grid knot of latitude φ , longitude θ , along the corresponding path *i*:

$$\delta \overline{C_i}(\varphi, \theta) = \int_{\varphi} \int_{\theta} K_i(\varphi, \theta) \delta C(\varphi, \theta) d\theta d\varphi,$$
(2)

where index *i* refers to the path for which the anomaly is computed and the kernels $K_i(\varphi, \theta)$ define the sensitivity area of the Rayleigh waveforms, regulating the weight of each grid point along the path [Lebedev and van der Hilst, 2008]. The dependence on ω is implied.

For each period, the local phase velocity anomaly at each grid knot, $\delta C(\varphi, \theta)$, can be decomposed into isotropic and anisotropic contributions, following a Fourier series expansion, in terms of the azimuth, ψ , of the wave number vector [*Smith and Dahlen*, 1973; *Deschamps et al.*, 2008].

$$\delta C(\psi) = \delta C_{iso} + A_{2\psi} \cos(2\psi) + B_{2\psi} \sin(2\psi) + A_{4\psi} \cos(4\psi) + B_{4\psi} \sin(4\psi), \tag{3}$$

where the latitude (φ), longitude (θ), and period ($2\pi/\omega$) dependence is assumed. The *A* and *B* terms can further be combined to estimate the amplitude (Λ) and the angle of the fast *S* wave polarization (Θ), for both 2ψ and 4ψ contributions (see Appendix B). These are useful to visualize the anisotropy patterns across the model.

Equation (2) is then discretized, and a linear system of equations (one equation for each path, Appendix B) is built [*Deschamps et al.*, 2008]. The system of equations is then solved independently at each frequency, using an LSQR algorithm (least squares with QR factorization [*Paige and Saunders*, 1982]), which does not provide a covariance matrix that can be used to quantify uncertainties. However, the fit to data may be expressed as the variance reduction of the model at each frequency/period, and resolution and leakage tests evaluate further the inversion's ability to recover seismic structure (section 3.2.2).

3.2.1. Model Parametrization, Regularization, and Data Fit

The model was first parametrized with a triangular grid [*Wang and Dahlen*, 1995] with an approximately constant interknot spacing of 40 km. The spacing was chosen to be sufficiently small for an accurate integration of the Rayleigh wave sensitivity kernels, $K(\varphi, \theta)$ [e.g., *Lebedev and van der Hilst*, 2008]. A larger triangular model grid was then constructed, with a 200 km interknot spacing similar to the average interstation distance across the network.

The LSQR method [*Paige and Saunders*, 1982] used to solve the linear system of equations requires two types of damping. Gradient damping penalizes the first derivatives of the phase velocity anomaly distribution. Double derivative damping, or "smoothing," moderates the second spatial derivatives, spreading out local anomaly peaks in the final velocity distribution [*Darbyshire and Lebedev*, 2009].

After visually inspecting different model outputs, for different damping and smoothing values, we chose the following combination for the isotropic, 2ψ and 4ψ velocity distributions, respectively: 0.05, 0.1, and 0.1 for damping and 0.5, 0.7, and 0.7 for smoothing, at all periods (see supporting information for more information). **3.2.2. Resolution and Leakage Tests**

To estimate the resolution of the phase velocity distribution, we built pseudocheckerboard models (Figure 3), calculated synthetic dispersion data, and inverted the resulting dispersion curves to recover the synthetic model. The pseudocheckerboard tests are constructed by setting the velocity anomaly at model grid points to $\pm 2.4\%$ and interpolating the values on the dense integration grid. Since the model grid used in this kind of tomography is based on a triangular geometry, instead of the classic Cartesian framework [e.g., *Nolet*, 1993; *Pawlak et al.*, 2011; *Boyce et al.*, 2016], the anomaly shapes in Figure 3 are in fact the closest to a "standard" checkerboard. This type of synthetic test was successfully used in previous models [e.g., *Deschamps et al.*, 2008; *Darbyshire and Lebedev*, 2009; *Legendre et al.*, 2014]. The width of the negative anomalies that we test for ranges from 200 to 400 km. The ~200 km wide anomalies were recovered only in the central and southwest parts of the model (Figure 3). The ~300 km anomalies were relatively well recovered throughout the model space (within a $\pm 1\%$ error), except in the extreme north and northeast of the region. The ~400 km anomaly



Figure 3. Resolution tests for isotropic low velocity checkerboard anomalies at period 100 s. The peak amplitude of the anomalies is $\pm 2.4\%$. (top row) The input model and (bottom row) the recovered model.

features were well recovered except in the extreme northeast, southeast, and southwest (Figure 3). We also note a ~0.5% increase in the recovered anomaly amplitude south of Hudson Bay, at the westernmost edge of our model (Figure 3).

Imperfect amplitude recovery may be due to poor resolution at the model edges, sparse interstation path coverage, and low degree of path crosscutting, resulting in directional sampling bias. For this reason, we manually shadowed the edges of the model and the areas that we consider unreliable by simultaneously visualizing resolution and leakage tests and the path distribution at each period. We thus focused our interpretations on features that are within the resolution capability of the area. Smoothing was also found to affect resolution by decreasing the peak amplitude and spatially spreading out anomaly features in the distribution [e.g., *Darbyshire and Lebedev*, 2009]. The number of paths (189–276), directions, and degree of crosscutting vary slightly in the 30–220 s period range (Figure S1 in the supporting information). A more detailed discussion of path coverage and resolution at short periods can be found in the supporting information [*Meier et al.*, 2004; *Deschamps et al.*, 2008]. Resolution tests performed at longer periods provided similar solutions as for shorter periods (e.g., 100 s, Figure 3), demonstrating that the change in path coverage/crosscutting, the main control on the resolution, does not impact significantly enough to justify an increase in smoothing/damping.

Numerical leakage between the inverted isotropic and anisotropic velocity components can introduce false velocity anomalies during the inversion procedure. To explore the impact of such numerical artifacts, we performed three separate resolution tests, in which we inverted each component individually: (1) purely isotropic (to investigate leakage into anisotropy), (2) purely anisotropic (to investigate leakage into isotropy), and (3) a combination that includes rotated anisotropy. Figure 4 shows the recovery of these synthetic components and the kind of numerical artifacts that appeared on the other components, at 100 s and 180 s periods.

In all three cases, the magnitudes of falsely induced isotropic or anisotropic anomalies were similar for the two periods. In the purely isotropic Earth model case, a mean anisotropy anomaly magnitude of 0.3% was



2 Ψ anisotropy

Isotropic velocity and rotated 2Ψ anisotropy



Figure 4. Three types of leakage tests performed at periods of 100 s and 180 s. The isotropic leakage test comprises the synthetic purely isotropic Earth model inputs, as recovered by the initial full inversion, and the corresponding output models at periods (a, b) 100 s and (c, d) 180 s, respectively. The anisotropic leakage test comprises the synthetic purely anisotropic Earth model inputs, as recovered by the initial full inversion, and the corresponding output models at periods of (e, f) 100 s and (g, h) 180 s, respectively. The third test comprises the isotropic Earth models, as recovered by the initial full inversion, with anisotropy orientations rotated by 90°, and the corresponding output models at periods of (i, j) 100 s and (k, l) 180 s, respectively.

10.1002/2016JB013599



Figure 5. Rayleigh wave isotropic phase velocity anomaly and azimuthal anisotropy maps of eastern Canada at periods ranging from 20 to 110 s. Vref is the regional reference velocity in km/s for each period. Phase velocity results are plotted within \pm 50 km from any ray path. Shadowed areas represent model edges that we deemed unreliable after simultaneously considering leakage, resolution tests, and path coverage at each period. Bottom right diagram shows the Rayleigh wave depth sensitivity kernels [*Lebedev and van der Hilst*, 2008].

leaked into the 2ψ component, with some model edge regions exhibiting false anisotropy anomalies up to 1% (northwestern area). Isotropic velocity patterns were well recovered with slightly reduced amplitudes (-0.3%) in parts of the Superior Craton at 100 s and increased values (+0.3%) at 180 s. In the purely anisotropic Earth model case (either 2ψ or 4ψ), there was little leakage into the isotropic component (mean isotropic phase



Figure 6. Rayleigh wave isotropic phase velocity anomaly and azimuthal anisotropy maps of eastern Canada at periods ranging from 120 to 220 s. Vref is the regional reference velocity in km/s for each period. Phase velocity results are plotted within \pm 50 km from any ray path. Shadowed areas represent model edges that we deemed unreliable after simultaneously considering leakage, resolution tests, and path coverage at each period. Bottom right diagram shows the Rayleigh wave depth sensitivity kernels [*Lebedev and van der Hilst*, 2008].

velocity leakage of 0.08%). The azimuthal anisotropy amplitudes and directions were well recovered, with an average anomaly reduction of ~0.2% and a mean azimuthal deviation from the input values of ~15°, mainly resulting from differences at the model edges. In the third test we inverted for an Earth model that comprises a combination of both isotropic and anisotropic components at 100 s and 180 s period. The anisotropy



Figure 7. (a) The variation of anisotropy magnitude with period at selected locations in the Superior Craton; UL, upper lithosphere; LL, lower lithosphere; TZ, transitional zone. (b) Standard deviation of phase velocity distribution at selected locations in the Superior Craton. (c) Sample locations for the selected regions in the Superior Craton. (d) Mean Rayleigh wave phase velocity dispersion at selected locations in the Superior Craton (bold dark grey line), mean regional phase velocity from our model (thin grey line), and synthetic dispersion curve calculated from the iasp91 global reference model (light grey line).

directions were rotated by 90°, to test for bias in the azimuth of the estimated *S* wave polarization. Figure 4 shows that the rotated anisotropy directions were well recovered, with high azimuthal deviations (> 60°) occurring in the northeast and the southernmost areas of the model. Together, these resolution tests provide further assurance that the inversion method as applied to our data set is able to generate credible tomographic models without significantly distorting each component or inducing significant numerical artifacts that could bias the interpretation.

4. Results

Figures 5 and 6 show the distribution of anisotropic Rayleigh wave phase velocities across eastern Canada for periods 20-220 s, with respect to the model average for each period. A first-order feature that dominates the model at periods <140 s is the strong difference between the seismically fast western section, which comprises provinces older than 2.5 Ga and the slow eastern section, comprising younger provinces.

The Superior Craton is seismically fast at periods ≤ 140 s, compared to adjacent provinces, and exhibits multiple distinct zones of anisotropy. Phase velocity anomalies range between ~0.5% near the Proterozoic border between the Superior Craton and the Grenville Province and ~2.5% locally to the SE of Hudson Bay at periods sampling mostly crustal and uppermost mantle depths (Figure 5). The local high-velocity maxima relocates toward the center of the province at intermediate periods (60–100 s). The 20 s map shows consistently similar isotropic velocity features with maps at periods of 35 s, and similar anisotropy magnitude and directions in the Grenville Province. Azimuthal anisotropy magnitude ranges between 0.5% and 1% in the 20–100 s period interval. Anisotropy directions at short periods (20–40 s) are not distributed in a coherent fashion, although an interesting pattern becomes increasingly evident in the 40–110 s period interval (peak depth sensitivity between 50 and 150 km, Figure 5): fast velocity axes gradually rotate from predominantly E-W in the south to N-S in the northern Superior section.

In the 110–140 s period interval, surface waves mostly sensitive to 150–200 km depth range sample upper mantle material that appears to be faster than the regional average ($dC/C \simeq 1\%$) and more homogenous

(the mean standard deviation of absolute phase velocity values is <25 m/s). The anisotropy directions are less coherent (Figure 6), and the mean anisotropy magnitude decreases by $\sim0.5\%$ (Figure 7a), indicating weakly anisotropic material beneath the province. The observed sudden drop in anisotropy anomaly strength is higher than twice the mean standard deviation in the distribution of anisotropy anomaly values estimated in the 110–140 s period interval in the Superior Province ($\sim0.1\%$, Figure 7a), making it a robust feature of our model.

The 140–180 s period range reveals a highly heterogeneous ($dC/C \simeq 3\%$) and anisotropic (~1.5%) mantle zone. At periods >180 s, the anisotropy increases significantly to 4% and strong heterogeneities are present ($dC/C \simeq -1.5\%$ to 2.5%).

The Grenville Province is characterized by the transition from the positive seismic anomaly that dominates the Archean Superior Craton to its west to the negative anomaly that underlies the Appalachian Province to its southeast (Figure 5). Wave speed anomaly amplitudes vary between -1.5% near the Appalachian Front and the Great Lakes and $\leq 1\%$ close to the Archean Superior boundary. Anisotropy amplitudes range between 0.2% and 1% and span a wide range of directions. Proterozoic provinces are ~2% slower than the cratonic mantle yet faster than the global average (Figure 5).

The Appalachian Province is underlain by consistently low isotropic phase velocities ($dC/C \simeq -1.5\%$ to -3.5%) with respect to the regional average.

5. Discussion

5.1. The Heterogeneous Mantle Beneath Precambrian Terranes 5.1.1. Comparison With Previous Studies

Most global [e.g., *Nettles and Dziewonski*, 2008; *Lebedev and van der Hilst*, 2008; *Ritsema et al.*, 2011; *Debayle and Ricard*, 2012; *Schaeffer and Lebedev*, 2013] and continental scale seismic tomography studies [e.g., van der Lee and Frederiksen, 2005; *Bedle and van der Lee*, 2009; *Yuan et al.*, 2011; *Schaeffer and Lebedev*, 2014] generally image a wide high-velocity anomaly beneath Precambrian North America, similar to our model (Figures 5 and 6). The large-scale velocity decrease beneath the Proterozoic Grenville lithosphere in our model (Figures 5 and 6) is also present in some global models [e.g., *Shapiro and Ritzwoller*, 2002; *Nettles and Dziewonski*, 2008; *Ritsema et al.*, 2011]. In contrast, some continental scale models [e.g., *van der Lee and Frederiksen*, 2005; *Bedle and van der Lee*, 2009; *Yuan et al.*, 2011; *Schaeffer and Lebedev*, 2014] image uniformly fast wave speeds beneath both Archean and Proterozoic terranes. A number of regional models of southeast Canada [*Chen and Li*, 2012; *Rondenay et al.*, 2000a; *Aktas and Eaton*, 2006] reported a high-velocity anomaly corridor dipping away from the Superior Craton beneath the Grenville Province at 50–150 km depth.

Beneath the Eastern Churchill Province, the decrease in velocity in our model correlates with the models of *Nettles and Dziewonski* [2008], *van der Lee and Frederiksen* [2005], *Bedle and van der Lee* [2009], and *Yuan et al.* [2011], while high-velocity anomalies are detected by *Schaeffer and Lebedev* [2014], *Lebedev and van der Hilst* [2008], and *Debayle and Ricard* [2012]. Previous tomographic models that included anisotropy in the inversion [*Nettles and Dziewonski*, 2008; *Yuan et al.*, 2011] correlate better with our model.

5.1.2. Seismic Heterogeneities Within the Superior Craton

The high seismic velocity generally imaged beneath cratonic regions is commonly thought to be caused by extensive chemical depletion in heavy elements such as Fe, Al, or Ca [e.g., *Jordan*, 1988; *Griffin et al.*, 2003]. Purely compositional variations ranging from highly depleted dunite, generally associated with Archean mantle, to typical modern mantle peridotites result in 1–2% velocity variations [e.g., *Goes et al.*, 2000; *Lee*, 2003; *James et al.*, 2004]. A higher degree of depletion in heavy metals is generally associated with lower density and higher velocity [e.g., *Griffin et al.*, 2009]. Some studies, however, suggest that the effects of melt depletion may be too subtle ($\delta V_s \leq 0.5\%$) to be resolved seismically [e.g., *Afonso et al.*, 2008; *Schutt and Lesher*, 2006]. The phase velocity anomalies beneath the Superior Province range between 0.5% and 2.5% at shorter periods (20–90 s) and –1% to 2.5% at longer periods (≥ 150 s), indicating strong lateral heterogeneities within the cratonic mantle. Studies of xenoliths from the Renard kimberlite field in central Quebec showed a 200 km thick lithosphere composed mainly of lherzolite and harzburgite [*Hunt et al.*, 2012], although more depleted harzburgite may be present, as well as eclogitic portions [*Hunt et al.*, 2013]. Variations in chemical

composition of xenoliths, indicative of mantle heterogeneities, have been reported in other Precambrian regions [e.g., Siberian craton, *Snyder et al.*, 1995, North China Craton, *Tang et al.*, 2013, and Slave Craton, *Kopylova and Caro*, 2004].

A 100 K temperature anomaly may cause a velocity change of 0.5–2% [e.g., *Hieronymus and Goes*, 2010]. Within the stable continental interiors, large temperature variations are unlikely, however, since the thermal equilibration times (~2 Ga for lithospheric thickness of 200-270 km [Michaut et al., 2007]) are similar to the time elapsed since the last tectonic event (~2.5 Ga in the Superior Craton). The most recent tectonic event that affected parts of the southern Superior Craton is the passage of the North American plate over the Great Meteor hot spot track at ~190-110 Ma ago (Figure 1). The hot spot may have refertilized the mantle [e.g., Frederiksen et al., 2013; Boyce et al., 2016] in the southern section of the study region, consistent with the negative velocity anomalies we observe at all periods in our model (Figure 5) also detected in some regional [e.g., Frederiksen et al., 2007; Villemaire et al., 2012; Boyce et al., 2016] and continental scale models [e.g., French et al., 2013]. The thermal impact associated with a mantle plume head can be as high as ~500°C, which will rapidly diffuse and decay to ~80°C at 200 km over 120 Ma [Eaton and Frederiksen, 2007]. A major thermal contribution to the lateral velocity variations within the upper lithosphere is thus unlikely. Seismic heterogeneities at periods sampling deeper mantle material may result from local lithospheric thickness variations. At the same period and depth range of propagation, Rayleigh waves may encounter low-velocity asthenospheric material, due to the local topographic relief of the cratonic root, resulting in strong lateral phase velocity heterogeneities at longer periods (>150 s).

5.1.3. Velocity Decrease Beneath Proterozoic and Younger Provinces

Within the Precambrian region, the gradual decrease in velocity (\leq 3%) is a large-scale feature of our model that persists to ~140 s period and broadly correlates with the Proterozoic terranes that surround the Superior Craton. The decrease in velocity from Archean to Proterozoic mantle may reflect the transition from a melt-depleted cratonic lithosphere to a more fertile modern mantle beneath the Proterozoic. Interestingly, this gradual change in mantle seismic properties with age mirrors a change in crustal properties: the post-Archean crust was recently found to be thicker, more heterogeneous, and more compositionally evolved [*Petrescu et al.*, 2016]. A recent body wave tomographic study of the upper mantle [*Boyce et al.*, 2016] detected a sharp vertical change in *P* and *S* wave velocity beneath the Grenville Front (Figure 1), which they interpreted as a metasomatic front, likely related to the pre-Grenvillian subduction system altering the composition of the cratonic margin.

A thermal effect linked with lithospheric thickness may also contribute to the phase velocity decrease, although it cannot be unequivocally discriminated from compositional effects. The conductive geothermal gradient of a thick lithosphere, generally imaged beneath cratons, can be associated with a decrease in $V_{\rm c}$ from 4.7 km/s in the upper lithosphere to 4.65 km/s at ~180 km depth, without considering anelastic attenuation [James et al., 2004]. Superior Craton lithospheric thickness was previously inferred to be >220 km [e.g., Yuan and Romanowicz, 2010; Schaeffer and Lebedev, 2014; Porritt et al., 2015], which would induce colder geotherms and consequently higher velocities. A thinner lithosphere that likely underlies the Proterozoic Grenville Province will have a steeper geothermal gradient, resulting in higher temperatures and progressively lower seismic velocities. The eastern continental margin of the Laurentian continent (present-day Grenville Province) is also thought to have been involved in a ~500 Ma long Andean-style paleosubduction system [Rivers, 1997; Rondenay et al., 2000a]; the Grenville Province was dominated by a hot long-duration plateau, similar to Tibet, that subsequently collapsed ~1 Ga ago [Rivers, 2009]. These tectonic events likely left an imprint on the cratonic mantle, altering its composition and lithospheric thickness and consequently redefining the geothermal gradient. We thus conclude that the joint thermochemical effect of both metasomatic refertilization [e.g., Boyce et al., 2016] of the Archean Laurentian mantle partially underlying the Grenville Province and a progressively thinner lithosphere can explain the ~3% velocity decrease we detect beneath the Proterozoic regions.

The ~5% velocity contrast between the Precambrian (i.e., the Superior, Eastern Churchill, and Grenville Provinces) and Phanerozoic terranes may arise from a thermal contrast: the Phanerozoic Appalachian Province may have experienced lithospheric delamination [e.g., *Levin and Park*, 2000; *Yuan and Levin*, 2014], which would have been accompanied by upwelling of hot asthenospheric material. Additionally, a shallower lithosphere underlain by hydrated asthenospheric melts beneath the Appalachians may also be contributing to the observed decrease in seismic velocity [e.g., *Rychert et al.*, 2005].



Figure 8. Rayleigh wave 2 ψ anisotropy at 100 s period (red lines). Black lines are surface tectonic boundaries. Green lines show previous SKS splitting measurements [*Sénéchal et al.*, 1996; *Rondenay et al.*, 2000b; *Eaton et al.*, 2004; *Frederiksen et al.*, 2006; *Darbyshire et al.*, 2015; *Gilligan et al.*, 2016]. Arrows are the APM vectors in the no-net-rotation frame [*DeMets et al.*, 2010] (dark blue) and in the hot spot frame [*Gripp and Gordon*, 2002] (light blue).

5.2. Causes of Upper Mantle Anisotropy

The main cause of anisotropy in the upper mantle is usually associated with olivine, the most abundant and highly anisotropic mineral of the mantle. Strain results in development of crystallographic preferred orientation (CPO) of olivine crystals, leading to a axis alignment in the direction of flow [e.g., Zhang and Karato, 1995; Bystricky et al., 2000; Tommasi et al., 2000]. When a shear wave traverses such a medium, the faster-traveling component will travel parallel to the a axes of the crystal [e.g., Silver and Chan, 1991]. Anisotropic fabrics can form due to asthenospheric flow [e.g., Vinnik et al., 1989; Fouch et al., 2000; Sleep et al., 2002], the preferential alignment of fluid or melt [e.g., Blackman and Kendall, 1997], preexisting anisotropy frozen into the lithosphere [e.g., Silver and Chan, 1988; Vauchez and Nicolas, 1991; Bastow et al., 2007], or any combination thereof.

5.3. Frozen-in Archean Tectonic Deformation in the Upper Lithosphere

Whether or not tectonic processes in the Archean were capable of imparting a seismic anisotropic fabric on the lithosphere is debated [e.g., *Fouch et al.*, 2004], but younger Precambrian fabrics are recognized worldwide. For example, the thick lithosphere of northern Canada retains a strong fossil lithospheric fabric of the ~1.8 Ga Trans-Hudson Orogen [*Bastow et al.*, 2011; *Darbyshire et al.*, 2013]. Previous measurements of anisotropy are concentrated in the southern sections of our study area and come from magnetotelluric [e.g., *Mareschal et al.*, 1995] and SKS measurements [e.g., *Darbyshire et al.*, 2015; *Gilligan et al.*, 2016], which have low vertical resolution and may often record a vertically averaged signal from the lithosphere and the convecting mantle.

At periods <110 s (peak sensitivity at ~120 km depth), anisotropy patterns within the Superior Craton transition from predominantly E-W in the south to N-S in the north (Figure 5), mimicking the orientation of the Archean surface boundaries (Figure 8). SKS splitting directions in the southern Superior are also parallel to the tectonic fabrics [*Sénéchal et al.*, 1996], although *Darbyshire et al.* [2015] found this correlation to be partial. Since the peak depth sensitivity of the maximum period at which this anisotropic signal is detected is at ~120 km depth, the source must reside in the lithosphere. Short scale lateral variation of SKS measurements also suggests that the origin of the anisotropic signal lies mostly in the 50–150 km depth range [*Sénéchal et al.*, 1996].

The striking agreement in direction between surface geological features, our calculated anisotropy in the 40–110 s period range (Figure 5), and previous SKS measurements (Figure 8) suggests that the anisotropic signal must be due to deeply rooted lithospheric deformation zones. This idea is also supported by a systematic obliquity between electric and seismic anisotropy identified in the southern Superior Craton [*Mareschal et al.*, 1995; *Sénéchal et al.*, 1996], thought to arise from the alignment of a-type olivine minerals in noncoaxial shear deformation zones in the lithosphere [*Ji et al.*, 1996]. The surface geology indicates that tectonic assembly of the Superior Craton was completed by 2.5 Ga [*Hoffman*, 1989], implying that the detected lithospheric deformation is of Archean age. The persistence of fossilized Archean tectonic deformation at mantle depths

implies that the crust and lithospheric mantle deformed coherently and that the constituent microcratonic blocks must have been already thick (\leq 120 km) and strong enough to transmit stress across lithospheric depth scales, at the time of Archean assemblage.

5.4. Archean Mantle Beneath Surrounding Terranes

Phase velocity anomalies of \sim 1%, indicative of cratonic material, partially extend beneath the Grenville Province. Anisotropic fabrics are obligue to the general NE-SW Grenville orogenic trend and parallel Archean deformation zones in the Superior. This observation is most evident at periods \leq 70 s, between 46° and 52°N (Figure 5), where the dominant E-W anisotropy direction persists up to 200 km away from the Archean-Proterozoic boundary. In the same area, SKS splitting measurements at a seismic array crossing the Archean-Proterozoic boundary showed no variation in the fast axis direction [Sénéchal et al., 1996; Ji et al., 1996]. Active seismic profiling of the Archean to Proterozoic crust detected Archean lower crust underlying the Grenville Province as far as \sim 250 km away from the Archean-Proterozoic boundary [*White et al.*, 2000]. Magnetotelluric studies detected dipping resistors to 120 km depth beneath the Grenville Province, interpreted as refertilized Archean mantle [e.g., Adetunji et al., 2014]. On the other hand, a recent body wave tomographic study of the upper mantle [Boyce et al., 2016] detected a sharp vertical change in P and S wave velocity beneath the Grenville Front (Figure 1). The joint thermochemical effect of metasomatism or a progressively thinner and warmer lithosphere, as discussed previously, may influence the slight reduction in seismic velocity beneath the circumcratonic regions. However, the persistence of anisotropic fabrics that parallel Archean surface deformation suggests that cratonic mantle extends beneath surrounding regions and maintains frozen-in Archean fabrics despite possible metasomatic reworking.

5.5. Anisotropy Decrease in the Lower Cratonic Lithosphere

We believe that the nature of the weakly anisotropic zone sampled in the 110-140 s period interval (peak sensitivity at 150-200 km depth) is lithospheric. Previous seismic [e.g., *Yuan and Romanowicz*, 2010; *Schaeffer and Lebedev*, 2014] and geochemical studies [*Hunt et al.*, 2012] inferred the lithospheric thickness in the area to be >220 km which is ~50 km deeper than the peak depth sensitivity at 140 s (Figure 6). Also, phase velocities are ~1% faster than the regional (~4.3 km/s, Figure 6) and the global average (~4.2 km/s, Figure 7d), indicating that the material is either colder or more compositionally depleted, akin to cratonic regions.

Competing effects of multiple anisotropic layers with perpendicular fast directions may cause an almost null cumulative anisotropic signature, which may be observed in SKS studies [e.g., *Girardin and Farra*, 1998]. Similarly, the relatively broad range of depth sensitivity of Rayleigh waves at periods between 110 and 140 s (Figure 6) may include weak signals from both the upper lithosphere and the underlying mantle, with contrasting anisotropic signatures, resulting in an apparently isotropic transition zone. However, if this was true, the observed splitting of SKS waves, which traveled through both anisotropic layers with hypothetically perpendicular anisotropy directions, should be almost null, contrary to previous measurements in the area [*Darbyshire et al.*, 2015; *Sénéchal et al.*, 1996]. Also, the plate motion vectors (Figure 8), commonly considered to represent upper mantle flow [e.g., *Yuan and Romanowicz*, 2010], are oblique to the anisotropy directions estimated in the upper lithosphere (20–140 s, Figure 5). We thus consider it unlikely that the interference between the anisotropic signatures of the upper lithosphere and the underlying flowing asthenosphere is sufficient to cause the observed drop in the absolute anisotropy magnitude in a period interval sampling a ~ 100 km depth range.

5.5.1. Hypotheses of Formation and Age of the Lower Lithosphere

Theories explaining the reduction in anisotropy within the upper mantle include a change in the creeping mechanism from dislocation to diffusion creep [*Karato*, 1992] or the olivine glide system [*Mainprice et al.*, 2005], both of which would imply a younger formation age for the lower lithosphere than for the upper lithosphere. Forward modeling of seismic anisotropy generated by olivine CPO developed in the presence of mantle flow suggests that a sudden switch in flow direction generates a depth-averaged response of the mineralogical CPO development, resulting in a weakly anisotropic basal layer at the transition between different strain orientations [*Tommasi*, 1998]. However, in a global study of seismic anisotropy, *Debayle and Ricard* [2013] found that for plate velocities of <4 cm/yr (current absolute plate motion (APM) magnitude in eastern Canada is ~20 mm/yr), the basal drag from plate-asthenosphere interaction is likely too weak to develop flow-aligned anisotropy. The absence of significant seismic anisotropy in the lower lithosphere may therefore suggest a low relative velocity between the growing Superior Craton lithosphere and the convecting mantle beneath during Precambrian times. The formation of the lower lithosphere may have been sufficiently protracted that no single, coherent mantle flow or basal drag anisotropic fabric was preserved within it.

Alternative explanations for the presence of weakly anisotropic layers include metasomatism by water-rich fluids or melts [e.g., Baptiste et al., 2012]. Annealed microstructures of coarse-grained peridotites extracted from ~150 km depth from the Kaapvaal Craton also indicate that dislocation creep, recording early Archean deformation, was followed by effective annealing, due to a protracted period of cooling and metasomatism [Baptiste et al., 2012]. Melt percolation derived from a volatile-rich subcontinental mantle may have reworked a preexisting layer from beneath, possibly erasing its original seismic signature. However, previous studies of mantle seismic anisotropy in Spain [Vauchez et al., 2005] and Polynesia [Tommasi et al., 2004] suggested that fossil tectonic fabrics, in the form of preexisting olivine crystal preferred orientations, inherited from older deformation, can survive partial melting or melt percolation. Baptiste et al. [2012] also speculated that metasomatism would occur episodically, which would induce lateral heterogeneities within affected areas. The Superior Craton is pervaded by multiple dyke swarms that record magmatic pulses at 2.5, 2.2, and 2.1 Ga, linked with episodes of extension at the paleocratonic margins [Ernst and Bleeker, 2010] or the arrival of mantle plume heads [Ernst and Bleeker, 2001], providing possible metasomatic sources. Geochemical data in the Slave Craton [Griffin et al., 2004] suggest the presence of a second uniform Archean mantle layer, with the top located between 140 and 160 km depth. Griffin et al. [2004] interpreted this layer as subcreted mantle plume material at ca. 3.2 Ga. However, the radial spread of plume material from one or more sources or the presence of pervasive dyke swarms in the lower lithosphere may be expected to cause discernable seismic anisotropy patterns [Yuan and Romanowicz, 2010; Snyder and Bruneton, 2007] and would decrease seismic velocities relative to normal lithospheric mantle [Tommasi et al., 2004], contrary to our results.

We speculate the timing of formation of the lower lithosphere to be Proterozoic, although our data alone cannot constrain this. While most geochemical evidence indicates that most melt extraction occurred in the Archean [*Carlson et al.*, 2005; *Pearson*, 1999], some xenoliths record Proterozoic mantle addition to the cratonic root [e.g., *Pearson et al.*, 1995; *Smart et al.*, 2014], supporting the idea that root formation and stabilization may be a younger phenomenon than the bulk depleted craton age. The wide extent of the almost isotropic and relatively homogenous lower lithosphere likely fits the hypothesis in which the base of the lithosphere grew by progressive cooling after the tectonic assemblage of the upper lithosphere was completed [e.g., *Bédard and Harris*, 2014]. The recognition of at least two temporally distinct lithospheric zones in our model endorses the idea that cratonic keel growth is not due to one particular dominant tectonic process and, most importantly, is not exclusively an Archean phenomenon. Previous seismic studies postulated a Paleoproterozoic origin for the lower layer beneath Hudson Bay and surrounding terranes [e.g., *Darbyshire et al.*, 2013; *Porritt et al.*, 2015], supporting episodic formation of cratonic roots in the Precambrian.

5.6. Lowermost Lithosphere — Transition to the Convective Asthenosphere

While we do not attempt to constrain the lithosphere-asthenosphere boundary due to the broad sensitivity range of Rayleigh waves at longer periods, we note that pronounced seismic heterogeneities develop at periods \geq 140 s and the anisotropy magnitude significantly increases at 190 s (~4%, Figure 6). Geodynamic models predict the existence of a thin transitional layer (\leq 50 km thick) at the base of the chemically depleted cratonic core, which could absorb some of the shear deformation [e.g., *Cooper et al.*, 2004]. The heterogeneous anisotropic material sampled between 140 and 190 s may be related to the geodynamically predicted transitional layer at the base of the keel.

At periods \geq 190 s, with peak depth sensitivity at ~270 km depth (Figure 6) high phase velocity anisotropy and heterogeneity develops. The data fit at these periods decreases by ~30% compared to shorter periods (Figure C1), which calls for caution when interpreting these features. Since we independently invert for phase velocity anomalies at each period, the consistency of large-scale features between periods suggests that these features are not random numerical artifacts but may indicate real material properties. Also, Figure 2 shows that velocity dispersion is clearly visible at frequencies as low as ~0.003 Hz (~333 s period). In accordance with our results, strong anisotropy is expected in the 250–400 km depth range beneath cratons, due to asthenospheric flow [*Gung et al.*, 2003]. Mantle convection and locally complex flow around the cratonic root may thus be responsible for the anisotropy increase and locally strong heterogeneities at periods \geq 190 s (Figure 6), corroborating the view that subcratonic mantle flow is complex [e.g., *Fouch et al.*, 2000].

5.7. Comparison With Other Cratons

Seismic anisotropy has been studied in other cratons using a range of methods such as SKS splitting [e.g., *Silver et al.*, 2001; *Snyder and Bruneton*, 2007], Love and Rayleigh wave anisotropy [e.g., *Adam and Lebedev*, 2012; *Yuan et al.*, 2014], and *P* and *S* receiver functions [e.g., *Abt et al.*, 2010; *Miller and Eaton*, 2010; *Sodoudi et al.*, 2013], each reporting slightly different results and interpretations, depending on the imaging method and

resolution. A comparison of the seismic anisotropy between different stable continental interiors yields a few similarities (see, e.g., *Fouch and Rondenay* [2006] for a review of anisotropy beneath cratons), although a unified anisotropic model of craton structure and growth seems increasingly unlikely. Some cratons are found to be anisotropic from top to bottom and may retain the signature of underplated lithosphere [e.g., Slave, *Bostock et al.*, 2010], while others may have been affected by nearby plume magmatism [e.g., Tanzania, *Sebai et al.*, 2006] or other types of thermochemical erosion (e.g., delamination beneath the North China Craton [*Xu*, 2001; *Gao et al.*, 2009]), which erased their initial anisotropic signatures or contributed to their growth or destruction in different ways. Most studies, however, agree that on average, anisotropy measurements display a correlation with surface tectonic trends within the upper ~150 km of the lithosphere, related to initial craton amalgamation, similar to our findings (Figure 8): the Baltic shield [e.g., *Pedersen*, 2006], the South African cratons [e.g., *Silver et al.*, 2004; *Fouch et al.*, 2004; *Adam and Lebedev*, 2012; *Freybourger et al.*, 2001], western Australian cratons [e.g., *Girardin and Farra*, 1998; *Fichtner et al.*, 2010], the North China Craton [e.g., *Bai et al.*, 2010], and the Slave Craton [e.g., *Snyder et al.*, 2003].

Most studies report a switch from dominant fossil anisotropy in the upper lithosphere to anisotropy related to present-day asthenospheric flow [e.g., *Deschamps et al.*, 2008]. The transition from layers containing frozen-in Archean deformation signatures to subcratonic flow may, however, be more complex. Mechanically weak basal layers such as deformable boundary layers that absorb some of the shear stress at the base of the keel have been geodynamically inferred [e.g., *Cooper et al.*, 2004] and reported beneath some cratons [e.g., *Kaban et al.*, 2015]. Also, lithospheric regions exhibiting a reduction in their azimuthal anisotropy similar to our observations (Figure 7) were previously reported in the 100–200 km depth range in a 3-D anisotropic model of North America [*Yuan et al.*, 2011] and between 70 and 150 km depth in Australia from full waveform tomography [*Fichtner et al.*, 2010]. Forward modeling of *S* receiver functions in the Kaapvaal Craton revealed an anisotropic layer (3% anisotropy) between 40 and 90 km depth, underlain by a dominantly isotropic meta-somatized lower lithosphere [*Sodoudi et al.*, 2013]. Thus, depending on the tectonic history of each craton, cratonic roots may comprise chemical and metasomatic layers [e.g., *Sodoudi et al.*, 2013], sometimes separated by intralithospheric discontinuities [e.g., *Wirth and Long*, 2014].

6. Conclusion

To investigate the structure of the lithosphere beneath previously unstudied parts of the Canadian Shield, we analyzed surface wave dispersion data from teleseismic earthquakes. We inverted Rayleigh wave dispersion data for isotropic phase velocity heterogeneity and azimuthal anisotropy variations at different frequencies.

A first-order feature of our model is the clear distinction in the isotropic phase velocity between the fast upper mantle underlying Archean terranes and that beneath Proterozoic belts, between 20 and 140 s. We interpret this contrast as a compositional change between relict depleted cratonic lithosphere and metasomatized Archean mantle extended beneath Proterozoic lithosphere.

The Superior Craton lithosphere is anisotropically complex. The upper lithosphere, sampled at periods of 40-100 s, is seismically fast and weakly heterogeneous, with anisotropic fabrics that parallel surface tectonic boundaries of Archean age. This correlation suggests that cratonic blocks must have been strong enough to support plate-scale collision at \sim 2.5 Ga. Cratonic upper lithosphere partially underlies the surrounding Proterozoic belts and preserves the signature of Archean deformation despite subsequent mantle refertilization. The lower lithosphere, sampled at periods of 110-140 s, is seismically fast, more homogenous, and weakly anisotropic. We speculate that it formed in Proterozoic times, subsequent to the accretion of the overlying Archean terranes. The weak anisotropy detected at lower lithospheric depths could result from the interference between present-day anisotropy in the flowing asthenosphere below and the fossil anisotropy in the upper lithosphere or the gradual transition between different strain orientations as a result of changing convection patterns at the time of lithospheric growth. Nevertheless, we find it more likely that the decrease represents the weakly anisotropic fabric of the material, which may point toward regionally sluggish mantle convection and/or slow absolute plate motion at the time of formation. Finally, the root of the cratonic mantle may be imaged between periods of 140 and 190 s and contains highly heterogeneous and anisotropic seismic fabrics. Significant topography may be present at the base of the craton, and the transition to the flowing mantle is therefore likely to be complex. Eastern Canada's heterogeneous and anisotropically complex lithosphere documents the development of cratonic keels through Archean and Proterozoic times.

Appendix A: Measurement Bias in the Two-Station Method

Individual dispersion curves measured between pairs of seismic stations describe the average medium properties between the two stations, which can be affected by the presence of strong lateral heterogeneity. Measuring the dispersion curves from multiple events from opposite directions enables the detection of such heterogeneities and improves the measurement quality [*Legendre et al.*, 2014]. Laterally heterogeneous media also generate seismic wave diffraction, introducing nonplane wave energy. Surface wave diffraction arises from an inherently complex relationship between waves and heterogeneities [*Pedersen*, 2006] and is not taken into account by the method used here. This phenomenon is manifested as irregularities in the dispersion curves (Figure 2). We used 10–250 measurements for each station pair (Figure 1) with seismic waves coming from opposite directions (Figure 2). These were sufficient to identify and exclude outlier dispersion curves, reducing the effects of heterogeneity and diffraction [*Meier et al.*, 2004; *Lebedev et al.*, 2006], and resulting in smooth dispersion curves (Figure S2 in the supporting information).

Bias may arise from angular deviations of the surface wave propagation paths from their predicted great-circle arc [*Pedersen*, 2006], with deviations of $\leq 20^{\circ}$ possible [e.g., *Cotte et al.*, 2000; *Foster et al.*, 2014]. At short periods (~20 s) great-circle arc deviations may be as high as 15° [*Pedersen et al.*, 2015]. *Foster et al.* [2014] showed that the standard deviation of the distribution of great-circle arc deviations decreases from 0.4 at short periods (~20 s) to less than 0.1 at longer periods (~100 s). *Pedersen* [2006] estimated that phase velocities can be $\leq 0.5\%$ different from the structural velocities for a deviation of 10° and 2–2.5% for a deviation of 20°. *Meier et al.* [2004] calculated that a deviation of 15° amounts to a phase velocity increase by a factor of 1.035. To reduce the effect of great-circle path deviations, we imposed a $\pm 5^{\circ}$ limit between the station pair azimuth and the earthquake azimuth, which should introduce an error <0.5% [*Pedersen*, 2006]. Also, averaging over hundreds of dispersion curves obtained from earthquake waves coming from opposite directions should significantly reduce the impact on phase velocity from wrongly predicted arrival angles for a few events.

Lateral scattering strongly influences fundamental mode Rayleigh waves at periods \leq 16 s [*Meier et al.*, 2004]. The JWKB approximation we use is insufficient to describe the complex wave propagation at such short periods. This is reflected in the incoherent pattern observed in the spectrograms above ~0.07 Hz (Figure 2). For this reason we do not use short-period data.

Appendix B: Inversion Setup

For each earthquake ray path connecting pairs of seismic stations, equation (2) is discretized. A linear system of equations is thus built and independently solved for at each period [*Deschamps et al.*, 2008]:

$$\boldsymbol{d} = \boldsymbol{G}\boldsymbol{m},\tag{B1a}$$

$$\boldsymbol{d}^{T} = \left(\overline{\delta}C_{1}\dots\overline{\delta}C_{N}\right),\tag{B1b}$$

$$\boldsymbol{m}^{\mathsf{T}} = \left(\delta C_{\mathrm{iso},1} \dots \delta C_{\mathrm{iso},M} A_{2\psi,1} \dots A_{2\psi,M} B_{2\psi,1} \dots B_{2\psi,M} A_{4\psi,1} \dots A_{4\psi,M} B_{4\psi,1} \dots B_{4\psi,M}\right), \tag{B1c}$$

where *N* is the number of paths obtained at each period and *M* is the number of grid knots.

The *A* and *B* terms are then combined to estimate the amplitude (Λ) and the angle of the fast *S* wave polarization (Θ), for both 2ψ and 4ψ contributions. These are useful to visualize the anisotropy patterns across the model (Figures 5, 6).

$$\Lambda = \sqrt{(A^2 + B^2)},\tag{B2a}$$

$$\Theta = \frac{1}{2}\arctan\left(\frac{B}{A}\right).$$
 (B2b)

Appendix C: The Importance of Anisotropy

Previous studies noted the importance of accounting for anisotropy in inverting and interpreting seismic data [e.g., *Hess*, 1964; *Forsyth*, 1973; *Deschamps et al.*, 2008; *Darbyshire and Lebedev*, 2009]. To assess how much the solution deteriorates if we neglect anisotropy, we calculated the variance reduction of the inverted models at all periods with (1) a purely isotropic inversion, (2) an inversion that includes isotropic heterogeneity and 2ψ anisotropic variation, and (3) the full inversion which allows for isotropic variations as well as 2ψ and 4ψ anisotropy. Figure C1 shows that the data fit decreases up to 25% when a purely isotropic solution is forced



Figure C1. (a) The variance reduction (data fit) decreases with period and increases with the number of terms (isotropic, 2ψ , and 4ψ) accounted for in the inversion. (b) The roughness of the isotropic, 2ψ , and 4ψ components, respectively, resulting from a full inversion, increases with period.

on the inversion versus a full solution. The phase velocity patterns are qualitatively similar if we only invert for the isotropic component (Figure S6 in the supporting information). However, the magnitude of the isotropic V_s anomaly estimated by an isotropic-only inversion may increase or decrease up to 2% in areas where the anisotropy estimated by the full inversion is $\geq 2\%$ or at the model edges, where path coverage and crosscutting is poor. The velocity heterogeneity can thus be overestimated, leading to biased interpretations.

Anisotropic olivine crystals in the mantle are thought to have only a small 4ψ signal [Montagner and Nataf, 1986], but the silica-rich nature of the continental crust [e.g., Christensen, 1996] means that it may have different anisotropic properties, rendering both the 2ψ and 4ψ anisotropy terms important [e.g., Polat et al., 2012]. Calculations of anisotropy for mineralogical and petrological upper mantle models indicate that Rayleigh waves have weak 4ψ anisotropy dependence, as opposed to Love waves [Montagner and Nataf, 1986; Montagner and Anderson, 1989]. Spatial coherence between 2ψ and 4ψ terms in a study of the Hellenic subduction zone found weak directional changes in the 2ψ terms, if the 4ψ component is not accounted for in the inversion [*Endrun et al.*, 2011]. They also found a correlation between higher 4ψ anomalies and regions with poorer ray path coverage. Our results, however, do not show such a correlation (see the supporting information), indicating that 4ψ terms are not simply an indication of fitted noise or a numerical strategy to improve data fit in less resolved model regions. The roughness of this component, defined as the average difference between the value at each grid knot and the nearest neighbor values, is also found to increase at longer periods (Figure C1b). Interpreting the 4ψ anisotropy from a physical point of view is not straightforward: we speculate tentatively that it may be a combination of Rayleigh-Love wave coupling effects or more complex mineralogical anisotropy systems developed locally. We therefore follow the practice of previous studies that calculate both 2ψ and 4ψ terms [e.g., Deschamps et al., 2008; Darbyshire and Lebedev, 2009; Fry et al., 2010; Polat et al., 2012] but focus our interpretation solely on the isotropic and the 2ψ anomalies, in the context of upper mantle structure.

Acknowledgments

The seismic data used in this study are freely available from the CNDC (Canadian National Data Centre for Earthquake Seismology and Nuclear Explosion Monitoring) and IRIS DMC (Data Management Center) via their data request tools. The Leverhulme Trust (grant RPG-2013-332) and National Science Foundation are acknowledged for financial support. L.P. is supported by Janet Watson Imperial College Department Scholarship and the Romanian Government Research Grant NUCLEU. F.D. is supported by NSERC through the Discovery Grants and Canada Research Chairs program. We also thank two anonymous reviewers and the Associate Editor for insightful comments that helped improve the manuscript.

References

Abt, D., K. Fischer, S. French, H. Ford, H. Yuan, and B. Romanowicz (2010), North American lithospheric discontinuity structure imaged by *Ps* and *Sp* receiver functions, *J. Geophys. Res.*, 115, B09301, doi:10.1029/2009JB006914.

Adam, J. M.-C., and S. Lebedev (2012), Azimuthal anisotropy beneath southern Africa from very broad-band surface-wave dispersion measurements, *Geophys. J. Int.*, 191(1), 155–174, doi:10.1111/j.1365-246X.2012.05583.x.

Adetunji, A. Q., I. J. Ferguson, and A. G. Jones (2014), Crustal and lithospheric scale structures of the Precambrian Superior-Grenville margin, Tectonophysics, 614, 146–169, doi:10.1016/j.tecto.2013.12.008.

Afonso, J. C., M. Fernández, G. Ranalli, W. Griffin, and J. Connolly (2008), Integrated geophysical-petrological modeling of the lithosphere and sublithospheric upper mantle: Methodology and applications, *Geochem. Geophys. Geosyst.*, 9, Q05008, doi:10.1029/2007GC001834.

Aktas, K., and D. W. Eaton (2006), Upper-mantle velocity structure of the lower Great Lakes region, *Tectonophysics*, 420(1–2), 267–281, doi:10.1016/j.tecto.2006.01.020.

Arndt, N., N. Coltice, H. Helmstaedt, and M. Gregoire (2009), Origin of Archean subcontinental lithospheric mantle: Some petrological constraints, *Lithos*, 109(1), 61–71, doi:10.1016/j.lithos.2008.10.019.

Bai, L., H. Kawakatsu, and Y. Morita (2010), Two anisotropic layers in central orogenic belt of North China Craton, *Tectonophysics*, 494(1), 138–148, doi:10.1016/j.tecto.2010.09.002.

Baptiste, V., A. Tommasi, and S. Demouchy (2012), Deformation and hydration of the lithospheric mantle beneath the Kaapvaal Craton, South Africa, *Lithos*, 149, 31–50, doi:10.1016/j.lithos.2012.05.001.

Bastow, I., T. Owens, G. Helffrich, and J. Knapp (2007), Spatial and temporal constraints on sources of seismic anisotropy: Evidence from the Scottish highlands, *Geophys. Res. Lett.*, 34, L05305, doi:10.1029/2006GL028911.

Bastow, I., D. Thompson, J. Wookey, J. Kendall, G. Helffrich, D. Snyder, D. Eaton, and F. Darbyshire (2011), Precambrian plate tectonics: Seismic evidence from northern Hudson Bay, Canada, *Geology*, 39(1), 91–94, doi:10.1130/G31396.1.

Bédard, J. (2006), A catalytic delamination-driven model for coupled genesis of Archaean crust and sub-continental lithospheric mantle, *Geochim. Cosmochim. Acta*, 70(5), 1188–1214, doi:10.1016/j.gca.2005.11.008.

Bédard, J. H., and L. B. Harris (2014), Neoarchean disaggregation and reassembly of the Superior Craton, *Geology*, 42(11), 951–954, doi:10.1130/G35770.1.

Bedle, H., and S. van der Lee (2009), S velocity variations beneath North America, J. Geophys. Res., 114, B07308, doi:10.1029/2008JB005949. Bizzarro, M., and R. K. Stevenson (2003). Major element composition of the lithospheric mantle under the North Atlantic Craton:

Evidence from peridotite xenoliths of the Sarfartoq area, southwestern Greenland, Contrib. Mineral. Petrol., 146(2), 223–240, doi:10.1007/s00410-003-0499-1.

Blackman, D., and J.-M. Kendall (1997), Sensitivity of teleseismic body waves to mineral texture and melt in the mantle beneath a mid-ocean ridge, *Philos. Trans. R. Soc. A*, 355, 217–231, doi:10.1098/rsta.1997.0007.

Bostock, M., D. Eaton, and D. Snyder (2010), Teleseismic studies of the Canadian landmass: Lithoprobe and its legacy, Can. J. Earth Sci., 47(4), 445–461, doi:10.1139/E09-040.

Boyce, A., I. Bastow, F. Darbyshire, A. Ellwood, A. Gilligan, V. Levin, and W. Menke (2016), Subduction beneath Laurentia modified the North American Cratonic Edge: Evidence from *P* and *S* wave tomography, *J. Geophys. Res. Solid Earth*, *121*, 5013–5030, doi:10.1002/2016JB012838.

Boyd, F., N. Pokhilenko, D. Pearson, S. Mertzman, N. Sobolev, and L. Finger (1997), Composition of the Siberian cratonic mantle: Evidence from Udachnaya peridotite xenoliths, *Contrib. Mineral. Petrol.*, 128(2–3), 228–246, doi:10.1007/s004100050305.

Bystricky, M., K. Kunze, L. Burlini, and J.-P. Burg (2000), High shear strain of olivine aggregates: Rheological and seismic consequences, *Science*, 290(5496), 1564–1567, doi:10.1126/science.290.5496.1564.

Carlson, R. W., D. G. Pearson, and D. E. James (2005), Physical, chemical, and chronological characteristics of continental mantle, *Rev. Geophys.*, 43, RG1001, doi:10.1029/2004RG000156.

Chen, C.-W., and A. Li (2012), Shear wave structure in the Grenville Province beneath the lower Great Lakes region from Rayleigh wave tomography, J. Geophys. Res., 117, B01303, doi:10.1029/2011JB008536.

Christensen, N. (1996), Poisson's ratio and crustal seismology, J. Geophys. Res., 101(B2), 3139-3156, doi:10.1029/95JB03446.

Cooper, C., A. Lenardic, and L. Moresi (2004), The thermal structure of stable continental lithosphere within a dynamic mantle, *Earth Planet. Sci. Lett.*, 222(3), 807–817, doi:10.1016/j.epsl.2004.04.008.

Cotte, N., H. Pedersen, M. Campillo, V. Farra, and Y. Cansi (2000), Off-great-circle propagation of intermediate-period surface waves observed on a dense array in the French Alps, *Geophys. J. Int.*, 142(3), 825–840, doi:10.1046/j.1365-246X.2000.00187.x.

Darbyshire, F., and S. Lebedev (2009), Rayleigh wave phase-velocity heterogeneity and multilayered azimuthal anisotropy of the Superior Craton, Ontario, *Geophys. J. Int.*, 176(1), 215–234, doi:10.1111/j.1365-246X.2008.03982.x.

Darbyshire, F., D. Eaton, and I. Bastow (2013), Seismic imaging of the lithosphere beneath Hudson Bay: Episodic growth of the Laurentian mantle keel, *Earth Planet. Sci. Lett.*, 373, 179–193, doi:10.1016/j.epsl.2013.05.002.

Darbyshire, F., I. Bastow, A. Forte, T. Hobbs, A. Calvel, A. Gonzalez-Monteza, and B. Schow (2015), Variability and origin of seismic anisotropy across eastern Canada: Evidence from shear wave splitting measurements, J. Geophys. Res. Solid Earth, 120, 8404–8421, doi:10.1002/2015JB012228.

Debayle, E., and Y. Ricard (2012), A global shear velocity model of the upper mantle from fundamental and higher Rayleigh mode measurements, J. Geophys. Res., 117, B10308, doi:10.1029/2012JB009288.

Debayle, E., and Y. Ricard (2013), Seismic observations of large-scale deformation at the bottom of fast-moving plates, *Earth Planet. Sci. Lett.*, 376, 165–177, doi:10.1016/j.epsl.2013.06.025.

DeMets, C., R. G. Gordon, and D. F. Argus (2010), Geologically current plate motions, *Geophys. J. Int.*, 181(1), 1–80, doi:10.1111/j.1365-246X.2009.04491.x.

Deschamps, F., S. Lebedev, T. Meier, and J. Trampert (2008), Azimuthal anisotropy of Rayleigh-wave phase velocities in the east-central United States, *Geophys. J. Int.*, 173(3), 827–843, doi:10.1111/j.1365-246X.2008.03751.x.

Eaton, D., and A. Frederiksen (2007), Seismic evidence for convection-driven motion of the North American plate, *Nature*, 446, 428–431, doi:10.1038/nature05675.

Eaton, D., A. Frederiksen, and S.-K. Miong (2004), Shear-wave splitting observations in the lower Great Lakes region: Evidence for regional anisotropic domains and keel-modified asthenospheric flow, *Geophys. Res. Lett.*, 31, L07610, doi:10.1029/2004GL019438.

Endrun, B., S. Lebedev, T. Meier, C. Tirel, and W. Friederich (2011), Complex layered deformation within the Aegean crust and mantle revealed by seismic anisotropy, *Nat. Geosci.*, 4(3), 203–207, doi:10.1038/ngeo1065.

Ernst, R., and W. Bleeker (2001), Large mafic magmatic events through time and links to mantle-plume heads, GSA Spec. Pap., 352, 483–575, doi:10.1130/0-8137-2352-3.483.

Ernst, R., and W. Bleeker (2010), Large igneous provinces (LIPs), giant dyke swarms, and mantle plumes: Significance for breakup events within Canada and adjacent regions from 2.5 Ga to the present, *Can. J. Earth Sci.*, 47, 695–739, doi:10.1139/E10-025.

Fichtner, A., B. L. Kennett, H. Igel, and H.-P. Bunge (2010), Full waveform tomography for radially anisotropic structure: New insights into present and past states of the Australasian upper mantle, *Earth Planet. Sci. Lett.*, 290(3), 270–280, doi:10.1016/j.epsl.2009.12.003.

Forsyth, D. (1973), The structural evolution of an oceanic plate, *Geophys. J. Int.*, 35(1-3), 376-376, doi:10.1093/gji/35.1.376.

Foster, A., G. Ekström, and M. Nettles (2014), Surface wave phase velocities of the Western United States from a two-station method, *Geophys. J. Int.*, 196, 1189–1206, doi:10.1093/gjj/ggt454.

Fouch, M., A. Fischer, E. Parmentier, M. Wysession, and T. Clarke (2000), Shear wave splitting, continental keels, and patterns of mantle flow, J. Geophys. Res., 105, 6255–6276, doi:10.1029/1999JB900372.

Fouch, M., P. Silver, D. Bell, and J. Lee (2004), Small-scale variations in seismic anisotropy near Kimberley, South Africa, *Geophys. J. Int.*, 157, 764–774, doi:10.1111/j.1365-246X.2004.02234.x.

Fouch, M. J., and S. Rondenay (2006), Seismic anisotropy beneath stable continental interiors, *Phys. Earth Planet. Inter.*, 158(2), 292–320, doi:10.1016/j.pepi.2006.03.024.

Frederiksen, A., I. Ferguson, D. Eaton, S.-K. Miong, and E. Gowan (2006), Mantle fabric at multiple scales across an Archean – Proterozoic boundary, Grenville Front, Canada, Phys. Earth Planet. Inter., 158(2), 240–263, doi:10.1016/j.pepi.2006.03.025.

Frederiksen, A., S. Miong, F. Darbyshire, D. Eaton, S. Rondenay, and S. Sol (2007), Lithospheric variations across the Superior Province, Ontario, Canada: Evidence from tomography and shear wave splitting, J. Geophys. Res., 112, B07318, doi:10.1029/2006JB004861.

Frederiksen, A., T. Bollmann, F. Darbyshire, and S. van der Lee (2013), Modification of continental lithosphere by tectonic processes: A tomographic image of central North America, *J. Geophys. Res. Solid Earth*, *118*, 1051–1066, doi:10.1002/jgrb.50060.

French, S., V. Lekic, and B. Romanowicz (2013), Waveform tomography reveals channeled flow at the base of the oceanic asthenosphere, Science, 342(6155), 227–230, doi:10.1126/science.1241514.

Freybourger, M., J. Gaherty, and T. Jordan (2001), Structure of the Kaapvaal Craton from surface waves, Geophys. Res. Lett., 28(13), 2489–2492, doi:10.1029/2000GL012436.

Fry, B., F. Deschamps, E. Kissling, L. Stehly, and D. Giardini (2010), Layered azimuthal anisotropy of Rayleigh wave phase velocities in the European Alpine lithosphere inferred from ambient noise, *Earth Planet. Sci. Lett.*, 297(1), 95–102, doi:10.1016/j.epsl.2010.06.008.

Gao, S., J. Zhang, W. Xu, and Y. Liu (2009), Delamination and destruction of the North China Craton, Chin. Sci. Bull., 54(19), 3367–3378, doi:10.1007/s11434-009-0395-9.

Gilligan, A., et al. (2016), Lithospheric deformation in the Canadian Appalachians: Evidence from shear wave splitting, *Geophys. J. Int., 206*(2), 1273–1280, doi:10.1093/gji/ggw207.

Girardin, N., and V. Farra (1998), Azimuthal anisotropy in the upper mantle from observations of *P*-to-S converted phases: Application to southeast Australia, *Geophys. J. Int.*, 133(3), 615–629, doi:10.1046/j.1365-246X.1998.00525.x.

Goes, S., R. Govers, and P. Vacher (2000), Shallow mantle temperatures under Europe from P and S wave tomography, J. Geophys. Res., 105(B5), 11,153–11,169, doi:10.1029/1999JB900300.

Griffin, W., B. Doyle, C. Ryan, N. Pearson, S. O'Reilly, R. Davies, K. Kivi, E. Van Achterbergh, and L. Natapov (1999), Layered mantle lithosphere in the Lac de Gras area, Slave Craton: Composition, structure and origin, J. Petrol., 40(5), 705–727, doi:10.1093/petroj/40.5.705.

Griffin, W., S. O'Reilly, N. Abe, S. Aulbach, R. Davies, N. Pearson, B. Doyle, and K. Kivi (2003), The origin and evolution of Archean lithospheric mantle, *Precambrian Res.*, 127(1), 19–41, doi:10.1016/S0301-9268(03)00180-3.

Griffin, W., S. O'Reilly, B. Doyle, N. Pearson, H. Coopersmith, K. Kivi, V. Malkovets, and N. Pokhilenko (2004), Lithosphere mapping beneath the North American plate, *Lithos*, 77(1), 873–922, doi:10.1016/j.lithos.2004.03.034.

Griffin, W., S. O'Reilly, J. Afonso, and G. Begg (2009), The composition and evolution of lithospheric mantle: A re-evaluation and its tectonic implications, J. Petrol., 50(7), 1185–1204, doi:10.1093/petrology/egn033.

Gripp, A., and R. Gordon (2002), Young tracks of hotspots and current plate velocities, *Geophys. J. Int.*, 150(2), 321–361, doi:10.1046/j.1365-246X.2002.01627.x.

Gung, Y., M. Panning, and B. Romanowicz (2003), Global anisotropy and the thickness of continents, *Nature*, 422(6933), 707-711, doi:10.1038/nature01559.

Hess, H. (1964), Seismic anisotropy of the uppermost mantle under oceans, Nature, 203, 629-631, doi:10.1038/203629a0.

Hieronymus, C., and S. Goes (2010), Complex cratonic seismic structure from thermal models of the lithosphere: Effects of variations in deep radiogenic heating, *Geophys. J. Int.*, 180(3), 999–1012, doi:10.1111/j.1365-246X.2009.04478.x.

Hoffman, P. F. (1989), Precambrian geology and tectonic history of North America, in *The Geology of North America*, edited by A. W. Bally and A. R. Palmer, pp. 447–512, Geol. Soc. of Am., Boulder, Colo.

Hunt, L., T. Stachel, and J. Armstrong (2010), Xenoliths and xenocrysts from the Renard Kimberlites, Quebec: A Comprehensive study of mantle samples to determine the evolution of the Superior Craton, paper presented at 2010 Canadian Society of Exploration Geophysicists Convention, Calgary, Alberta, Canada, 10–14 May.

Hunt, L., T. Stachel, H. Grütter, J. Armstrong, T. E. McCandless, A. Simonetti, and S. Tappe (2012), Small mantle fragments from the Renard Kimberlites, Quebec: Powerful recorders of mantle lithosphere formation and modification beneath the eastern Superior Craton, J. Petrol., 53(8), 1597–1635, doi:10.1093/petrology/egs027.

James, D. E., F. Boyd, D. Schutt, D. Bell, and R. Carlson (2004), Xenolith constraints on seismic velocities in the upper mantle beneath southern Africa, *Geochem. Geophys. Geosyst.*, *5*, Q01002, doi:10.1029/2003GC000551.

Ji, S., S. Rondenay, M. Mareschal, and G. Senechal (1996), Obliquity between seismic and electrical anisotropies as a potential indicator of movement sense for ductile shear zones in the upper mantle, *Geology*, 24(11), 1033–1036, doi:10.1029/0004-0002-0004-0004-0002-0004-004-00

doi:10.1130/0091-7613(1996)024<1033:OBSAEA>2.3.CO;2.

Jordan, T. (1988), Structure and formation of the continental tectosphere, J. Petrol., 1, 11–37, doi:10.1093/petrology/Special_Volume1.11. (Special Lithosphere Issue).

Kaban, M. K., W. D. Mooney, and A. G. Petrunin (2015), Cratonic root beneath North America shifted by basal drag from the convecting mantle, *Nat. Geosci.*, 8(10), 797–800, doi:10.1038/ngeo2525.

Karato, S.-I. (1992), On the Lehmann discontinuity, *Geophys. Res. Lett.*, *19*(22), 2255–2258, doi:10.1029/92GL02603.

Karato, S.-I. (2012), On the origin of the asthenosphere, *Earth Planet. Sci. Lett.*, 321, 95–103, doi:10.1016/j.epsl.2012.01.001.Kopylova, M., and G. Caro (2004), Mantle xenoliths from the southeastern Slave Craton: Evidence for chemical zonation in a thick, cold lithosphere, *J. Petrol.*, 45(5), 1045–1067, doi:10.1093/petrology/egh003.

Kopylova, M. G., and J. K. Russell (2000), Chemical stratification of cratonic lithosphere: Constraints from the northern Slave Craton, Canada, Earth Planet. Sci. Lett., 181(1), 71–87, doi:10.1016/S0012-821X(00)00187-4.

Lebedev, S., and R. van der Hilst (2008), Global upper-mantle tomography with the automated multimode inversion of surface and S-wave forms, Geophys. J. Int., 173(2), 505–518, doi:10.1111/j.1365-246X.2008.03721.x.

Lebedev, S., T. Meier, and R. D. van der Hilst (2006), Asthenospheric flow and origin of volcanism in the Baikal Rift area, *Earth Planet. Sci. Lett.*, 249(3), 415–424, doi:10.1016/j.epsl.2006.07.007.

Lee, C.-T. A. (2003), Compositional variation of density and seismic velocities in natural peridotites at STP conditions: Implications for seismic imaging of compositional heterogeneities in the upper mantle, J. Geophys. Res., 108(B9), 2441, doi:10.1029/2003JB002413.

Legendre, C., F. Deschamps, L. Zhao, S. Lebedev, and Q.-F. Chen (2014), Anisotropic Rayleigh wave phase velocity maps of eastern China, J. Geophys. Res. Solid Earth, 119, 4802–4820, doi:10.1002/2013JB010781.

Levin, V., and J. Park (2000), Shear zones in the Proterozoic lithosphere of the Arabian Shield and the nature of the Hales discontinuity, *Tectonophysics*, 323(3–4), 131–148, doi:10.1016/S0040-1951(00)00105-0.

Li, C., R. van der Hilst, R. Engdahl, and S. Burdick (2008), A new global model for *P* wave speed variations in Earth's mantle, *Geochem. Geophys. Geosyst.*, 9, Q05018, doi:10.1029/2007GC001806.

Mainprice, D., A. Tommasi, H. Couvy, P. Cordier, and D. Frost (2005), Pressure sensitivity of olivine slip systems and seismic anisotropy of Earth's upper mantle, *Nature*, 433, 731–733, doi:10.1038/nature03266.

Mareschal, M., R. Kellett, R. Kurtz, J. Ludden, S. Ji, and R. Bailey (1995), Archaean cratonic roots, mantle shear zones and deep electrical anisotropy, *Nature*, 375, 134–137, doi:10.1038/375134a0.

Meier, T., K. Dietrich, B. Stöckhert, and H.-P. Harjes (2004), One-dimensional models of shear wave velocity for the eastern Mediterranean obtained from the inversion of Rayleigh wave phase velocities and tectonic implications, *Geophys. J. Int.*, 156(1), 45–58, doi:10.1111/j.1365-246X.2004.02121.x.

Michaut, C., C. Jaupart, and D. Bell (2007), Transient geotherms in Archean continental lithosphere: New constraints on thickness and heat production of the subcontinental lithospheric mantle, *J. Geophys. Res., 112*, B04408, doi:10.1029/2006JB004464.

Miller, M., and D. Eaton (2010), Formation of cratonic mantle keels by arc accretion: Evidence from S receiver functions, Geophys. Res. Lett., 37, L18305, doi:10.1029/2010GL044366.

Montagner, J.-P., and D. L. Anderson (1989), Petrological constraints on seismic anisotropy, *Phys. Earth Planet. Inter.*, 54(1-2), 82-105, doi:10.1016/0031-9201(89)90189-1.

Montagner, J.-P., and H.-C. Nataf (1986), A simple method for inverting the azimuthal anisotropy of surface waves, J. Geophys. Res., 91(B1), 511–520, doi:10.1029/JB091iB01p00511.

Nettles, M., and A. Dziewonski (2008), Radially anisotropic shear velocity structure of the upper mantle globally and beneath North America, J. Geophys. Res., 113, B02303, doi:10.1029/2006JB004819.

Nolet, G. (1993), Solving large linearized tomographic problems, in *Seismic Tomography: Theory and Practice*, edited by H. Iyer and K. Hirahara, pp. 227–247, Chapman and Hall, London.

Paige, C. C., and M. A. Saunders (1982), LSQR: An algorithm for sparse linear equations and sparse least squares, ACM Trans. Math. Software, 8(1), 43–71, doi:10.1145/355993.356000.

Pawlak, A., D. Eaton, I. Bastow, J.-M. Kendall, G. Helffrich, J. Wookey, and D. Snyder (2011), Crustal structure beneath Hudson Bay from ambient noise tomography: Implications for basin formation, *Geophys. J. Int.*, 184(1), 65–82, doi:10.1111/j.1365-246X.2010.04828.x. Pearson, D. (1999), The age of continental roots, *Lithos*, 48(1), 171–194, doi:10.1016/S0024-4937(99)00026-2.

Pearson, D., R. Carlson, S. Shirey, F. Boyd, and P. Nixon (1995), Stabilisation of Archaean lithospheric mantle: A Re Os isotope study of peridotite xenoliths from the Kaapvaal Craton, *Earth Planet. Sci. Lett.*, 134(3), 341–357, doi:10.1016/0012-821X(95)00125-V.

Pedersen, H., P. Boué, P. Poli, and A. Colombi (2015), Arrival angle anomalies of Rayleigh waves observed at a broadband array: A systematic study based on earthquake data, full waveform simulations and noise correlations, *Geophys. J. Int., 203*(3), 1626–1641, doi:10.1093/gji/ggv382.

Pedersen, H. A. (2006), Impacts of non-plane waves on two-station measurements of phase velocities, *Geophys. J. Int.*, 165(1), 279–287, doi:10.1111/j.1365-246X.2006.02893.x.

Petrescu, L., I. Bastow, F. Darbyshire, A. Gilligan, T. Bodin, W. Menke, and V. Levin (2016), Three billion years of crustal evolution in eastern Canada: Constraints from receiver functions, J. Geophys. Res. Solid Earth, 121, 788–811, doi:10.1002/2015JB012348.

Plomerová, J., and V. Babuska (2010), Long memory of mantle lithosphere fabric—European LAB constrained from seismic anisotropy, Lithos, 120(1), 131–143, doi:10.1016/j.lithos.2010.01.008.

Polat, G., S. Lebedev, P. Readman, B. O'Reilly, and F. Hauser (2012), Anisotropic Rayleigh-wave tomography of Ireland's crust: Implications for crustal accretion and evolution within the Caledonian Orogen, *Geophys. Res. Lett.*, 39, L04302, doi:10.1029/2012GL051014.

Porritt, R. W., M. S. Miller, and F. A. Darbyshire (2015), Lithospheric architecture beneath Hudson Bay, Geochem. Geophys. Geosyst., 16, 2262–2275, doi:10.1002/2015GC005845.

Ritsema, J., A. Deuss, H. van Heijst, and J. Woodhouse (2011), S40RTS: A degree-40 shear-velocity model for the mantle from new Rayleigh wave dispersion, teleseismic traveltime and normal-mode splitting function measurements, *Geophys. J. Int.*, 184(3), 1223–1236, doi:10.1111/j.1365-246X.2010.04884.x.

Rivers, T. (1997), Lithotectonic elements of the Grenville Province: Review and tectonic implications, *Precambrian Res.*, 86(3), 117–154, doi:10.1016/S0301-9268(97)00038-7.

Rivers, T. (2009), The Grenville Province as a large hot long-duration collisional orogen—Insights from the spatial and thermal evolution of its orogenic fronts, *Geol. Soc. Lond. Spec. Publ.*, 327(1), 405–444, doi:10.1144/SP327.17.

Romanowicz, B. (2009), The thickness of tectonic plates, Science, 324(5926), 474–476, doi:10.1126/science.1172879.

Rondenay, S., M. G. Bostock, T. M. Hearn, D. J. White, and R. M. Ellis (2000a), Lithospheric assembly and modification of the SE Canadian Shield: Abitibi-Grenville teleseismic experiment, *J. Geophys. Res.*, *105*(B6), 13,735–13,754, doi:10.1029/2000JB900022.

Rondenay, S., M. G. Bostock, T. M. Hearn, D. J. White, H. Wu, G. Sénéchal, S. Ji, and M. Mareschal (2000b), Teleseismic studies of the lithosphere below the Abitibi-Grenville Lithoprobe transect, *Can. J. Earth Sci.*, *37*(2–3), 415–426, doi:10.1139/e98-088.

Rychert, C. A., K. M. Fischer, and S. Rondenay (2005), A sharp lithosphere – asthenosphere boundary imaged beneath eastern North America, *Nature*, 436(7050), 542–545, doi:10.1038/nature03904.

Sato, Y. (1955), Analysis of dispersed surface waves by means of Fourier transform: Part 1, *Bull. Earthquake Res. Inst. Univ. Tokyo*, 33, 33–47. Savage, B., and P. G. Silver (2008), Evidence for a compositional boundary within the lithospheric mantle beneath the Kalahari Craton from S receiver functions, *Earth Planet. Sci. Lett.*, 272(3), 600–609, doi:10.1016/j.epsl.2008.05.026.

Schaeffer, A., and S. Lebedev (2013), Global shear speed structure of the upper mantle and transition zone, *Geophys. J. Int.*, 194(1), 417–449, doi:10.1093/gji/ggt095.

Schaeffer, A., and S. Lebedev (2014), Imaging the North American continent using waveform inversion of global and USArray data, Earth Planet. Sci. Lett., 402, 26–41, doi:10.1016/j.epsl.2014.05.014.

Schutt, D., and C. Lesher (2006), Effects of melt depletion on the density and seismic velocity of garnet and spinel lherzolite, J. Geophys. Res., 111, B05401, doi:10.1029/2003JB002950.

Sebai, A., E. Stutzmann, J. Montagner, D. Sicilia, and E. Beucler (2006), Anisotropic structure of the African upper mantle from Rayleigh and Love wave tomography, *Phys. Earth Planet. Inter.*, 155(1–2), 48–62, doi:10.1016/j.pepi.2005.09.009.

Sénéchal, G., S. Rondenay, M. Mareschal, J. Guilbert, and G. Poupinet (1996), Seismic and electrical anisotropies in the lithosphere across the Grenville Front, Canada, *Geophys. Res. Lett.*, 23(17), 2255–2258, doi:10.1029/96GL01410.

Shapiro, N., and M. Ritzwoller (2002), Monte-Carlo inversion for a global shear-velocity model of the crust and upper mantle, *Geophys. J. Int.*, 151(1), 88–105, doi:10.1046/j.1365-246X.2002.01742.x.

Silver, P., and W. Chan (1988), Implications for continental structure and evolution from seismic anisotropy, *Nature*, 335(6185), 34–39, doi:10.1038/335034a0.

Silver, P., and G. Chan (1991), Shear wave splitting and subcontinental mantle deformation, J. Geophys. Res., 96(B10), 16,429–16,454, doi:10.1029/91JB00899.

Silver, P., S. Gao, and K. Liu (2001), Mantle deformation beneath southern Africa, *Geophys. Res. Lett.*, 28, 2493–2496, doi:10.1029/2000GL012696.

Silver, P., M. Fouch, S. Gao, and M. Schmitz (2004), Seismic anisotropy, mantle fabric, and the magmatic evolution of Precambrian southern Africa, S. Afr. J. Geol., 107, 45–58, doi:10.2113/107.1-2.45.

Sleep, N., C. Ebinger, and J.-M. Kendall (2002), Deflection of mantle plume material by cratonic keels, J. Geol. Soc. Lond., 199, 135–150, doi:10.1144/GSL.SP.2002.199.01.08.

Sleep, N. H. (1990), Monteregian hotspot track: A long-lived mantle plume, J. Geophys. Res., 95(B13), 21,983–21,990, doi:10.1029/JB095iB13p21983.

- Smart, K. A., T. Chacko, A. Simonetti, Z. D. Sharp, and L. M. Heaman (2014), A record of Paleoproterozoic subduction preserved in the northern Slave cratonic mantle: Sr-Pb-O isotope and trace-element investigations of eclogite xenoliths from the Jericho and Muskox kimberlites, J. Petrol., 55, 549-583, doi:10.1093/petrology/egt077.
- Smith, M., and F. Dahlen (1973), Azimuthal dependence of Love and Rayleigh wave propagation in a slightly anisotropic medium, J. Geophys. Res., 78(17), 3321–3333, doi:10.1029/JB078i017p03321.
- Snieder, R., and G. Nolet (1987), Linearized scattering of surface-waves on a spherical Earth, J. Geophys., 61(1), 55-63.
- Snyder, D., and M. Bruneton (2007), Seismic anisotropy of the Slave Craton, NW Canada, from joint interpretation of SKS and Rayleigh waves, *Geophys. J. Int.*, 169(1), 170–188, doi:10.1111/j.1365-246X.2006.03287.x.
- Snyder, D., M. Bostock, and G. Lockhart (2003), Two anisotropic layers in the Slave Craton, *Lithos*, 71, 529–539, doi:10.1016/j.lithos.2003.09.001.
- Snyder, G. A., L. A. Taylor, E. A. Jerde, R. N. Clayton, T. K. Mayeda, P. Deines, G. R. Rossman, and N. V. Sobolev (1995), Archean mantle heterogeneity and the origin of diamondiferous eclogites, Siberia: Evidence from stable isotopes and hydroxyl in garnet, Am. Mineral., 80(7–8), 799–809, doi:10.2138/am-1995-7-820.
- Sodoudi, F., X. Yuan, R. Kind, S. Lebedev, J. M.-C. Adam, E. Kästle, and F. Tilmann (2013), Seismic evidence for stratification in composition and anisotropic fabric within the thick lithosphere of Kalahari Craton, *Geochem. Geophys. Geosyst.*, 14, 5393–5412, doi:10.1002/2013GC004955.
- Tang, Y.-J., H.-F. Zhang, J.-F. Ying, B.-X. Su, Z.-Y. Chu, Y. Xiao, and X.-M. Zhao (2013), Highly heterogeneous lithospheric mantle beneath the Central Zone of the North China Craton evolved from Archean mantle through diverse melt refertilization, *Gondwana Res.*, 23(1), 130–140, doi:10.1016/j.gr.2012.01.006.

Thybo, H. (2006), The heterogeneous upper mantle low velocity zone, *Tectonophysics*, *416*(1), 53–79, doi:10.1016/j.tecto.2005.11.021. Tommasi, A. (1998), Forward modeling of the development of seismic anisotropy in the upper mantle, *Earth Planet. Sci. Lett.*, *160*(1–2), 1–13. doi:10.1016/S0012-821X(98)00081-8.

Tommasi, A., D. Mainprice, G. Canova, and Y. Chastel (2000), Viscoplastic self-consistent and equilibrium-based modeling of olivine lattice preferred orientations: Implications for the upper mantle seismic anisotropy, J. Geophys. Res., 105(B4), 7893–7908, doi:10.1029/1999JB900411.

- Tommasi, A., M. Godard, G. Coromina, J. Dautria, and H. Barsczus (2004), Seismic anisotropy and compositionally induced velocity anomalies in the lithosphere above mantle plumes: A petrological and microstructural study of mantle xenoliths from French Polynesia, *Earth Planet. Sci. Lett.*, 227(3), 539–556, doi:10.1016/j.epsl.2004.09.019.
- van der Lee, S., and A. Frederiksen (2005), Surface wave tomography applied to the North American upper mantle, in Seismic Earth: Array Analysis of Broadband Seismograms, vol. 157, edited by A. Levander and G. Nolet, pp. 67–80, AGU, Washington, D. C.

van Staal, C., and S. Barr (2012), Lithospheric architecture and tectonic evolution of the Canadian Appalachians and associated Atlantic margin, in *Tectonic Styles in Canada: The LITHOPROBE Perspective, Geol. Assoc. of Can. Spec. Pap.*, vol. 49, edited by J. A. Percival, F. A. Cook, and R. M. Clowes, pp. 41–95, Geol. Assoc. of Can., St. John's, Newfoundland.

- Vauchez, A., and A. Nicolas (1991), Mountain building: Strike-parallel motion and mantle anisotropy, *Tectonophysics*, 185(3), 183–201, doi:10.1016/0040-1951(91)90443-V.
- Vauchez, A., F. Dineur, and R. Rudnick (2005), Microstructure, texture and seismic anisotropy of the lithospheric mantle: Insights from the Labait volcano xenoliths (Tanzania), *Earth Planet. Sci. Lett.*, 232, 295–314, doi:10.1016/j.epsl.2005.01.024.
- Villemaire, M., F. Darbyshire, and I. Bastow (2012), P-wave tomography of eastern North America: Evidence for mantle evolution from Archean to Phanerozoic, and modification during subsequent hotspot tectonism, J. Geophys. Res., 112, B12302, doi:10.1029/2012JB009639.
- Vinnik, L., V. Farra, and B. Romanowicz (1989), Azimuthal anisotropy in the Earth from observations of SKS at Geoscope and NARS broadband stations, Bull. Seismol. Soc. Am., 79(5), 1542–1558.
- Wang, Z., and F. Dahlen (1995), Spherical-spline parameterization of three-dimensional Earth models, *Geophys. Res. Lett.*, 22(22), 3099–3102, doi:10.1029/95GL03080.
- Wardle, R. J., B. Ryan, and I. Ermanovics (1990), The eastern Churchill province, Torngat and New Quebec orogens: An overview, *Geosci. Can.*, 17(4), 217–222.
- White, D., D. Forsyth, I. Asudeh, S. Carr, H. Wu, R. Easton, and R. Mereu (2000), A seismic-based cross-section of the Grenville Orogen in southern Ontario and western Quebec, Can. J. Earth Sci., 37(2–3), 183–192, doi:10.1139/e99-094.
- Wirth, E. A., and M. D. Long (2014), A contrast in anisotropy across mid-lithospheric discontinuities beneath the central United States A relic of craton formation, *Geology*, 42(10), 851–854, doi:10.1130/G35804.1.

Worthington, J. R., B. R. Hacker, and G. Zandt (2013), Distinguishing eclogite from peridotite: EBSD-based calculations of seismic velocities, *Geophys. J. Int.*, 193(1), 489–505, doi:10.1093/gjji/ggt004.

- Xu, Y.-G. (2001), Thermo-tectonic destruction of the Archaean lithospheric keel beneath the Sino-Korean Craton in China: Evidence, timing and mechanism, Phys. Chem. Earth, 26(9), 747–757, doi:10.1016/S1464-1895(01)00124-7.
- Yuan, H., and V. Levin (2014), Stratified seismic anisotropy and the lithosphere-asthenosphere boundary beneath eastern North America, J. Geophys. Res. Solid Earth, 119, 3096–3114, doi:10.1002/2013JB010785.
- Yuan, H., and B. Romanowicz (2010), Lithospheric layering in the North American Craton, *Nature*, 466, 1063–1068, doi:10.1038/nature09332. Yuan, H., B. Romanowicz, K. Fischer, and D. Abt (2011), 3-D shear wave radially and azimuthally anisotropic velocity model of the North
 - American upper mantle, Geophys. J. Int., 184(3), 1237–1260, doi:10.1111/j.1365-246.2010.04901.x.
- Yuan, H., S. French, P. Cupillard, and B. Romanowicz (2014), Lithospheric expression of geological units in central and eastern North America from full waveform tomography, *Earth Planet. Sci. Lett.*, *1*, 1–11, doi:10.1016/j.epsl.2013.11.057.

Zhang, S., and S.-I. Karato (1995), Lattice preferred orientation of olivine aggregates deformed in simple shear, *Nature*, 375, 774–777, doi:10.1038/375774a0.